\begin{thebibliography}{10}

\bibitem{AcharyaS96Study}
A.~Acharya and J.~Saltz.
\newblock {A Study of Internet Round-Trip Delay}.
\newblock Technical Report CS-TR-3736, University of Maryland, December 1996.

\bibitem{AcharyaS97Dynamic}
A.~Acharya and J.~Saltz.
\newblock {\em {Dynamic Linking for Mobile Programs}}, pages 245--62.
\newblock Springer Verlag, 1997.
\newblock {\em Mobile Object Systems}, Jan Vitek and Christian Tschudin (eds).

\bibitem{ACHARYA96}
A.~Acharya, M.~Uysal, R.~Bennett, A.~Mendelson, M.~Beynon, J.~K. Hollingsworth,
  J.~Saltz, and A.~Sussman.
\newblock Tuning the performance of {I/O} intensive parallel applications.
\newblock In {\em Proceedings of the Fourth Annual Workshop on I/O in Parallel
  and Distributed Systems (IOPADS)}, May 1996.

\bibitem{AdlTabatabaiLLW96}
A.~Adl-Tabatabai, G.~Langdale, S.~Lucco, and R.~Wahbe.
\newblock {Efficient and Language-Independent Mobile Programs}.
\newblock In {\em Proceedings of the SIGPLAN'96 Conference on Programming
  Language Design and Implementation}, pages 127--36, May 1996.

\bibitem{AmsalegFT96}
L.~Amsaleg, M.~Franklin, and A.~Tomasic.
\newblock {Query Scrambling for Bursty Data Arrival}.
\newblock Technical Report CS-TR-3714, Dept of Computer Science, University of
  Maryland, 1996.

\bibitem{AmsalegFTU96}
L.~Amsaleg, M.~Franklin, A.~Tomasic, and T.~Urhan.
\newblock {Scrambling Query Plans to Cope With Unexpected Delays}.
\newblock In {\em Proceedings of the Fourth International Conference on
  Parallel and Distributed Information Systems}, December 1996.

\bibitem{Borenstein94}
N.~Borenstein.
\newblock {Email With a Mind of its Own: The Safe-TCL Language for Enabled
  Mail}.
\newblock In {\em Proceedings of IFIP Working Group 6.5 International
  Conference}, pages 389--402, Jun 1994.

\bibitem{BadrinathW95}
{B.R.Badrinath and Girish Welling}.
\newblock {Event Delivery Abstractions for Mobile Computing}.
\newblock Technical Report LCSR-TR-242, {Rutgers University}, 1996.

\bibitem{NoblePS95}
{Brian D. Noble and Morgan Price and M.Satyanarayanan}.
\newblock {A Programming Interface for Application-Aware Adaptation in Mobile
  Computing}.
\newblock {\em {{Proceedings of the Second USENIX Symposium on Mobile and
  Location Independent Computing}}}, Feb. 1995.

\bibitem{Cardelli95}
L.~Cardelli.
\newblock {A Language With Distributed Scope}.
\newblock In {\em Proceedings of the 22nd ACM Symposium on Principles of
  Programming Languages}, Jan. 1995.

\bibitem{ClamenLNW90}
S.~Clamen, L.~Leibengood, S.~Nettles, and J.~Wing.
\newblock {Reliable Distributed Computing with Avalon/Common Lisp}.
\newblock In {\em {Proceedings of the International Conference on Computer
  Languages}}, pages {169--79}, 1990.

\bibitem{Falcone87}
J.~Falcone.
\newblock {A Programmable Interface Language for Heterogeneous Systems}.
\newblock {\em {ACM Transactions on Computer Systems}}, 5(4):{330--51}, Nov.
  1987.

\bibitem{GoslingM95}
J.~Gosling and H.~McGilton.
\newblock {The Java Language Environment White Paper}, 1995.

\bibitem{Gray96}
R.~Gray.
\newblock {Agent TCL: A Flexible and Secure Mobile-agent System}.
\newblock In {\em {Proceedings of the Fourth Annual Tcl/Tk Workshop (TCL 96)}},
  July 1996.

\bibitem{JohansenRS95}
D.~Johansen, R.~van Renesse, and F.~Schneider.
\newblock {An Introduction to the {TACOMA} Distributed System Version 1.0}.
\newblock Technical Report {95-23}, {University of Tromso}, 1995.

\bibitem{JulLHB88}
E.~Jul, H.~Levy, N.~Hutchinson, and A.~Black.
\newblock {Fine-Grained Mobility in the Emerald System}.
\newblock {\em {ACM Transactions on Computer Systems}}, 6(2):{109--33}, Feb.
  1988.

\bibitem{StamosG90}
{J.W. Stamos and D.K. Gifford}.
\newblock {Implementing Remote Evaluation}.
\newblock {\em {{IEEE Transactions on Software Engineering}}},
  16(7):{710--722}, July 1990.

\bibitem{Katz96}
R.~Katz.
\newblock {The Case for Wireless Overlay Networks}.
\newblock Invited talk at the ACM Federated Computer Science Research
  Conferences, Philadelphia, 1996.

\bibitem{eos}
M.~King.
\newblock {Earth Observation System Project Science} homepage.
\newblock {\em http://\-spso.gsfc.nasa.gov/\-spso\_homepage.html}, 1995.

\bibitem{LangeO96}
D.~Lange and M.~Oshima.
\newblock {\em {Programming Mobile Agents in Java}}.
\newblock In progress, 1996.
\newblock (ch 2,3).

\bibitem{MummertES95}
L.~Mummert, M.~Ebling, and M.~Satyanarayanan.
\newblock {Exploiting Weak Connectivity for Mobile File Access}.
\newblock In {\em {Proceedings of the Fifteenth ACM Symposium on Operating
  System Principles}}, Dec. {1995}.

\bibitem{StraberBH96}
M.~Stra{$\beta$}er, J.~Baumann, and F.~Hohl.
\newblock {Mole - A Java Based Mobile Agent System}.
\newblock In {\em Proceedings of the ECOOP'96 workshop on Mobile Object
  Systems}, 1996.

\bibitem{JavaRMI}
{Sun Microsystems}.
\newblock {\em Java Remote Method Invocation}.
\newblock {\em http://chatsubo.javasoft.com/\-current/\-rmi/\-index.html}.

\bibitem{White96}
J.~White.
\newblock {Telescript Technology: Mobile Agents}, 1996.
\newblock {\em http://www.genmagic.com\-/Telescript\-/Whitepapers}.

\end{thebibliography}

\section{Design constraints}
\label{sec:constraints}

In this section, we discuss the language design constraints that arise
from the desire to be resource-aware and the use of mobility as a
mechanism to adapt to changes in resource availability. We discuss the
requirements of awareness, agility and authority and the constraints
each of them generates.  In the next section, we describe Sumatra and
discuss how its design has been guided by these constraints.


\subsection{Awareness}
\label{sec:awareness}

Resource-aware programs need to be able to monitor the availability
and quality of the resources in their environment.  A resource can be
monitored either on-demand or continuously.  Both kinds of monitoring
are useful. On-demand monitoring is useful in three kinds of
situations. First, if the resource in question is used infrequently
but is expensive to use -- e.g. an application on a mobile host that
uses a cell-modem to periodically scan incoming mail being held at a
post-office machine. Second, if the availability of the resource in
question changes infrequently -- e.g. an application that chooses the
location from which it monitors a process based on the amount of disk
space available at that location. Third, if the resource is expensive
to monitor and the cost of monitoring outweighs the potential gains --
e.g. an application that accesses large volumes of data over a very
slow link. Continuous monitoring is useful if the resource is
frequently used or if the resource changes frequently or is cheap to
monitor.

On-demand monitoring is, by necessity, synchronous. The request for
information does not return until the information is available.
Continuous monitoring can, on the other hand, support both synchronous
and asynchronous interfaces. A synchronous request would return
immediately with information about the current availability of the
resource in question whereas an asynchronous request would monitor the
resource and notify the application only when the resource
availability no longer satisfies an application-specified predicate. A
synchronous interface is suitable, for example, for an application
that sends a sequence of large messages and checks the state of the
network before sending each message. An asynchronous interface is
useful, for example, for an application that uses a resource
continuously (like receiving a video stream over the network) or an
application that does not use the network itself but calls library
routines which may do so. An asynchronous interface is also useful to
inform applications about a qualitative change in the operating
environment, for example if the platform is mobile and may need to
switch between multiple wireless networks~\cite{Katz96}. 

For continuous monitoring, an important decision is the granularity at
which resource change is to be monitored. The simplest alternative is
to report every change to the requesting application. This is often
impractical as most resource levels have some jitter which usually has
little impact on application performance. The next simplest
alternative is to use a jitter threshold and track only those changes
that larger than this threshold. Jitter-threshold-based schemes have
been proposed by several researchers as a way of dealing with resource
variability on mobile platforms~\cite{BadrinathW95,NoblePS95}.
Jitter-threshold-based schemes work well if changes in the resource
levels are usually stable. Transient changes (usually just spikes) in
the resource levels can cause spurious responses. The alternative is
to augment the jitter-based scheme with a filter that eliminates
transients. This allows the applications to track only the stable
changes. It is therefore important to allow applications to register
application-specific (or resource-specific) filters that determine
which changes in resource availability/quality should be reported to
the requesting application.

To summarize, languages for resource-aware mobile programs should
provide a resource-monitoring interface that allows on-demand
monitoring as well as continuous monitoring. For continuous monitoring,
the resource-monitoring interface should allow programs to register a
application-specific filter which determines which resource changes
should be reported.

\subsection{Agility}
\label{sec:agility}

To achieve agility, a mobile code language should provide mechanisms
that allow programs to react quickly to asynchronous events like
revocation of allocated bandwidth/memory or qualitative changes in
network connectivity (e.g. on a mobile host). Two mechanisms are
required.  First, the ability to receive the event in an asynchronous
manner (a la Unix signals) and second, the ability to take appropriate
action in response to such events including moving program execution
to a different site.

The requirement that programs be able to move within event handling
code constrains the choice of mobility mechanisms.  There are two
major alternatives: (1) a {\tt go()} (or a {\tt jump}) primitive that
freezes execution at current site and resumes execution at target
site; and (2) a function-call-like mechanism which allows programs to
execute a procedure at a specified site. Most mobile code languages
have selected one of these two alternatives.  Agent-Tcl~\cite{Gray96},
Telescript~\cite{White96} and Aglets~\cite{LangeO96} use a {\tt
go}-based mechanism whereas Obliq~\cite{Cardelli95},
Avalon~\cite{ClamenLNW90}, NCL~\cite{Falcone87} REV~\cite{StamosG90}
and TACOMA~\cite{JohansenRS95} use a function-call-based interface.
To ensure a prompt response to asynchronous events, a
function-call-based interface would require one of two things: (1)
either the language automatically captures the continuation at the
point at which an event occurs and makes it available for use within
the corresponding event handler; or (2) the programmer emulates this
functionality by writing a large number of functions that represent
continuations at different points in the programs. In both cases, the
language has to allow the use of a continuation-passing style.  We
believe a {\tt go()}-based interface is simpler to use; the astute
reader has probably already noted that the first option has exactly
the same effect as a {\tt go()}.

\subsection{Authority}
\label{sec:authority}

The requirement of authority is, by far, the most demanding. It
requires that language allow programs to control the way in which
resources are used on their behalf by system support as well as by
libraries. In effect, it requires that module boundaries not be
completely opaque and that they allow resource-usage related
restrictions to pass through and enforced as a part of execution.

There is a trade-off between the extent to which programs are allowed
to control resource-use and ease of programming. At one extreme, a
language can require that operations that use resources of interest be
performed only after they have been explicitly authorized. This allows
complete control over resource usage. Unauthorized accesses would
raise an exception. A scheme similar to this is used by Java applets
to control access to local resources (see
Figure~\ref{fig:java-security} for an example). This works well for
Java applets as the level of access to local resources is not usually
changed during the execution of a program and there is no advantage in
retrying the operation.

On the other hand, it is entirely possible that the response to
authorization failure for resource-usage can be (and in many cases,
will be) to change the level of authorization. In such cases, retrying
the operation would be desirable. This can be a problem if
authorization failures occur deep in library code. It may or may not
be possible to restart the operation if the failure is delivered as an
exception. For example, consider the code in
Figure~\ref{fig:restart-problem}. For this example, assume that the
resource of interest is the number of sockets. If the first call to
checkCreateSocket() succeeds and the second fails (say the program has
already created as many sockets as it was permitted to), an exception
will be raised. A common response to this situation would be to
negotiate with the resource manager for a higher limit on the number
of sockets and to restart the operation. However, a clean restart for
{\tt createTwoSockets()} is not possible as the first socket has
already been created.

This situation can be dealt with by treating resource-use violation as
an asynchronous event and allowing programs to associate a handler
with every restriction on resource-use. The handler would be executed
in the same context as the operation that caused the violation and
would allow the operation to be restarted, if that is what is desired.

To summarize, both forms of control over resource use are desirable:
(1) checking all operations that might use resources of interest for
authorization and delivering an exception if the usage is not
authorized; (2) considering resource-use violation as an asynchronous
event and allowing programs to associate event handlers to
resource-use restrictions which execute in the same context as the
faulting operation.


\begin{figure}
\tightboxit{
\begin{tabbing}
ttttttttt\=ttt\=ttt\=ttt\= \kill
\>{\tt  public boolean mkdir() \{ \ \ \ \ \ \ \ \ \ \ \ \ \ \ }\\
\>\>{\tt        SecurityManager security = System.getSecurityManager();}\\
\>\>{\tt        if (security != null) \{}\\
\>\>\>{\tt            security.checkWrite(path);}\\
\>\>{\tt        \}}\\
\>\>{\tt        return mkdir0();}\\
\>{\tt    \}}\\
\\
\>{\tt    public boolean renameTo(File dest) \{}\\
\>\>{\tt        SecurityManager security = System.getSecurityManager();}\\
\>\>{\tt        if (security != null) \{}\\
\>\>\>{\tt            security.checkWrite(path);}\\
\>\>\>{\tt            security.checkWrite(dest.path);}\\
\>\>{\tt        \}}\\
\>\>{\tt        return renameTo0(dest);}\\
\>{\tt    \}}\\
\end{tabbing}
}
\caption{Excerpts from Java class libraries that illustrate how access
to local resources is controlled. The actual operations of creating 
a directory and renaming a file are performed by {\tt mkdir0()} and
{\tt renameTo0()}.}
\label{fig:java-security}
\end{figure}

\begin{figure}
\tightboxit{
\begin{tabbing}
tt\=ttt\=ttt\=ttt\= \kill
\>{\tt public boolean createSocket(String host, int port, boolean stream) \{}\\
\>\>{\tt        ResourceManager resource = System.getResourceManager();}\\
\>\>{\tt        if (resource != null) \{ }\\
\>\>\>{\tt        resource.checkCreateSocket(host,port,stream);}\\
\>\>{\tt        \}}\\
\>\>{\tt        return Socket(host,port,stream);}\\
\>{\tt    \}}\\
\\
\>{\tt  private Socket socket1;}\\
\>{\tt  private Socket socket2;}\\
\> .... \\
\>{\tt  public void createTwoSockets(String host1,String host2,int port,}\\
\>\>\>{\tt boolean stream) \{ }\\
\> ... \\
\>\>{\tt        socket1 = createSocket(host1,port,stream);}\\
\>\>{\tt        socket2 = createSocket(host2,port,stream);} {\bf // fails authorization check}\\
\>{\tt    \}}\\
\end{tabbing}
}
\caption{Sample Java code to illustrate the problem with 
restarting operations that fail authorization checks. This code assumes 
that Java is extended with a Resource Manager similar to its current
Security Manager.}
\label{fig:restart-problem}
\end{figure}

